Author's response to reviews

Title: Glutamine therapy reduces endothelial adhesion of sickle red blood cells to human umbilical vein endothelial cells

Authors:

Yutaka Niihara M.D. (ysniihara@msn.com)
Neil M Matsui Ph.D. (nmatsui@itsa.ucsf.edu)
Yamin M Shen M.D. (ysniihara@msn.com)
Dean A Akiyama (dakiyama@socal.rr.com)
Cage S Johnson M.D. (cagejohn@usc.edu)
Maria A Sunga M.D. (vicnor1@earthlink.net)
John Mapayo M.D. (imagpayo@aol.com)
Stephen H Embury M.D. (sembury@itsa.ucsf.edu)
Vijay K Kalra Ph.D. (ykalra@usc.edu)
Seong H Cho M.D. (praisecho@yahoo.com)
Kouichi R Tanaka M.D. (kthucla@aol.com)

Version: 8 Date: 14 July 2005

Author's response to reviews:

Title was changed to sentence case.

Country was added.

Acknowledgement was placed in the appropriate section.

Abstract was formatted as advised. Abstract in submission page was changed.

Manuscript section format was revised as advised.

References style was revised as advised.

Figures were double checked.

Figure legends were placed after the references.

List of abbreviations was added after conclusions

Competing interests and authors' contributions were listed.

Thank you.